\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{???.??} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{??X??} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ 39A12,39A13,34L10\ \mscs }
  }%\vbox
}




\begin{document}
\begin{thebibliography}{99}                                                                                               %
	
	
	\bibitem {ankitap}M. H. Annaby, Z. S. Mansour , \textit{Basic Sturm--Liouville
		problems}, J. Phys. A, Math Gen. 38, (17)\ $\left(  2005\right)  $\ 3775-3797.\ 
	
	\bibitem {32}B. P. Allahverdiev, H. Tuna, \textit{Uniform convergence of
		generalized Fourier series of Hahn--Sturm--Liouville problem}, Konuralp J.
	Math. 9 (2) (2021), 250-259.
	
	\bibitem {02}K. Aydemir, O. Sh. Mukhtarov, \textit{Class of Sturm--Liouville
		problems with eigenparameter dependent transmission conditions}, Numer. Funct.
	Analys. Optimiz. 38 (10) (2017), 1260-1275.
	
	\bibitem {ber}Ju. M. Berezanski\u{\i}, \textit{Expansions in Eigenfunctions of
		Selfadjoint Operators}, Amer. Math. Soc., Providence, 1968.
	
	\bibitem {er}T. Ernst, \textit{The History of }$q$\textit{-Calculus and a New
		Method}, U. U. D. M. Report (2000): 16, ISSN1101-3591, Department of
	Mathematics, Uppsala University, 2000.
	
	\bibitem {g2}G. Sh. Guseinov, \textit{Eigenfunction expansions for a
		Sturm--Liouville problem on time scales}, Int. J. Difference Equ. 2 (1)
	(2007), 93-104.
	
	\bibitem {han1}W. Hahn, \textit{\"{U}ber orthogonalpolynome, die }%
	$q$\textit{-Differenzengleichungen gen\"{u}gen}, Math. Nachr. 2 (1949), 4-34.
	
	\bibitem {kac}V. Kac, P. Cheung, \textit{Quantum Calculus}, Springer, New
	York, 2002.
	
	\bibitem {levinson}N. Levinson, \textit{A simplified proof of the expansion
		theorem for singular second order linear differential equations}, Duke Math.
	J. 18 (1951), 57-71.
	
	\bibitem {lev}B. M. Levitan, I. S. Sargsjan, \textit{Sturm--Liouville and
		Dirac Operators, Mathematics and its Applications} (Soviet Series), Kluwer
	Academic Publishers Group, Dordrecht, 1991 (translated from the Russian).
	
	\bibitem {nai}M. A. Naimark, \textit{Linear Differential Operators}, 2nd edn,
	Nauka, Moscow, 1969; English transl. of 1st edn, Parts 1, 2, Ungar, New York,
	1967, 1968.
	
	\bibitem {o12}H. Olgar, O. Sh. Mukhtarov, K. Aydemir, \textit{Some properties
		of eigenvalues and generalized eigenvectors of one boundary value problem},
	Filomat 32 (3) (2018), 911-920.
	
	\bibitem {stek}V.\ A. Steklov, \textit{Osnovnye Zadachi Matematicheskoi Fiziki
		(Basic problems of Mathematical Physics)}, 1, Petrograd, 1922.
	
	\bibitem {stone}M. H. Stone, \textit{A comparison of the series of Fourier and
		Birkhoff}, Trans. Amer. Math. Soc. 28 (1926), 695-761.
	
	\bibitem {tit}E. C. Titchmarsh, \textit{Eigenfunction Expansions Associated
		with Second-Order Differential Equation}s, Part I. Second Edition Clarendon
	Press, Oxford, 1962.
\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[Eigenfunction expansions of impulsive $q$-Sturm--Liouville problems]{\large Eigenfunction expansions of impulsive $q$-Sturm--Liouville problems}


\author[Bilender P. Allahverdiev]{{\bf Bilender P. Allahverdiev}\protect\orcidicon{0000-0002-9315-4652}}

\address{Bilender P. Allahverdiev  % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Department of Mathematics, Khazar University,
	\newline\hphantom{iii} AZ1096 Baku, Azerbaijan, and 
	\newline\hphantom{iii} Research Center of Econophysics,
	\newline\hphantom{iii} UNEC-Azerbaijan State University of Economics,
	\newline\hphantom{iii} Baku, Azerbaijan}%
\email{\textcolor{blue}{bilenderpasaoglu@gmail.com}}%

\author[H\"{u}seyin Tuna]{{\bf H\"{u}seyin Tuna}\protect\orcidicon{0000-0001-7240-8687}}

\address{H\"{u}seyin Tuna  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Department of Mathematics, Mehmet Akif Ersoy University,
\newline\hphantom{iii} 15030 Burdur, Turkey and,
\newline\hphantom{iii} Research Center of Econophysics,
\newline\hphantom{iii} UNEC-Azerbaijan State University of Economics,
\newline\hphantom{iii} Baku, Azerbaijan}%
\email{\textcolor{blue}{hustuna@gmail.com}}%

\author[Hamlet A. Isayev]{{\bf Hamlet A. Isayev}\protect\orcidicon{0000-0002-6383-0883}}

\address{Hamlet A. Isayev % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Department of Mathematics, Khazar University,
	\newline\hphantom{iii}  AZ1096 Baku, Azerbaijan}%
	\email{\textcolor{blue}{hamlet@khazar.org}}%
	

\thanks{}
\thanks{\copyright \ }
\thanks{\rm )}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc P.P. Petrov}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} In this work, an impulsive $q$-Sturm--Liouville problem is studied. The
existence of a countably infinite set of eigenvalues and eigenfunctions is
proved and a uniformly convergent expansion formula in the eigenfunctions is established.\medskip

\noindent{\bf Keywords:} Difference equations, impulsive conditions,
$q$-Sturm--Liouville problems, eigenfunction expansions.
 \end{quote}
}

\bigskip

\section{Introduction}
The Fourier method is one of the important
methods used in the solution of partial differential equations. While solving
with this method, the partial differential equation considered is reduced to
Sturm--Liouville problems. The eigenvalues and eigenfunctions of these
problems are investigated. Meanwhile, the problem of uniform convergence of
eigenfunction expansions arises. For this reason, eigenvalues, eigenfunctions,
and eigenfunction expansions of Sturm--Liouville problems have been
extensively studied in the theory of differential equations (see
\cite{ankitap,32,02,ber,g2,levinson,lev,nai,o12,stone,tit}).

Quantum calculus, which has a long history, has recently started to attract
the attention of many researchers ( see \cite{er}). In 2005 Annaby and Mansour
studied the $q$-analogue of the classical Sturm--Liouville equations (see
\cite{ankitap}). Classical results are obtained by considering the problem in
the regular case. Thereupon, the researchers began to study whether the
results obtained in the classical Sturm--Liouville theory were valid for the
$q$-Sturm--Liouville problems. In this context, the existence of countably
infinite sets of eigenvalues and eigenvectors of impulsive $q$%
-Sturm--Liouville problems and uniform convergence of eigenfunction expansions
will be investigated. The Uniform convergence problem will be investigated
using Steklov's method (see \cite{32,g2,stek}).

\section{Preliminaries}

In this section, the basic concepts of $q$-calculus that will be used in the
article will be given. For more detailed information, the following sources
can be examined,\  \cite{kac,ankitap,er}.

Let $q\in \left(  0,1\right)  \ $and let $A\subset \mathbb{R\ }$be a
$q$-geometric set, i.e., if $q\zeta \in A$ for all $\zeta \in A.$ We begin by
defining the \textit{operator} $\mathcal{D}_{q}$ by%
\[
\mathcal{D}_{q}f\left(  \zeta \right)  =\left \{
\begin{array}
	[c]{cc}%
	\frac{f\left(  q\zeta \right)  -f\left(  \zeta \right)  }{\left(  q-1\right)
		\zeta}, & \zeta \neq0\\
	\lim_{n\rightarrow \infty}\frac{f\left(  q^{n}\xi \right)  -f\left(  0\right)
	}{q^{n}\xi}, & \zeta=0,
\end{array}
\right.  \text{ }%
\]
where $\zeta,\xi \in A.\ $We define the \textit{Jackson} $q$%
-\textit{integration} by%
\[
\int_{0}^{\zeta}f\left(  \gamma \right)  d_{q}\gamma=\zeta \left(  1-q\right)
\sum_{n=0}^{\infty}q^{n}f\left(  q^{n}\zeta \right)  ,
\]
where $\zeta \in A.$ From \cite{han1}, we have%
\[
\int_{0}^{\infty}f\left(  \gamma \right)  d_{q}\gamma=\sum_{n=-\infty}^{\infty
}q^{n}f\left(  q^{n}\right)  .
\]
Through the remainder of the paper, we deal only with functions $q$-regular at
zero, i.e., functions satisfying%
\[
\lim_{n\rightarrow \infty}f\left(  \zeta q^{n}\right)  =f\left(  0\right)  ,
\]
for every $\zeta \in A.$

\section{Main Results}

Consider the following problem%
\begin{equation}
	(\tau y)\left(  \zeta \right)  :=-\frac{1}{q}D_{q^{-1}}D_{q}y\left(
	\zeta \right)  +v\left(  \zeta \right)  y\left(  \zeta \right)  =\lambda y\left(
	\zeta \right)  ,\  \zeta \in(0,d)\cup(d,a) \label{1.1}%
\end{equation}
with the boundary and impulsive conditions
\begin{equation}%
	\begin{tabular}
		[c]{l}%
		$y\left(  0\right)  -h_{1}D_{q^{-1}}y\left(  0\right)  =0,$\\
		$y\left(  a\right)  +h_{2}D_{q^{-1}}y\left(  a\right)  =0,$%
	\end{tabular}
	\  \  \label{1.3}%
\end{equation}%
\begin{align}
	y\left(  d-\right)   &  =\eta y\left(  d+\right)  ,\text{ }\label{i1}\\
	\mathcal{D}_{q^{-1}}y\left(  d-\right)   &  =\frac{1}{\eta}\mathcal{D}%
	_{q^{-1}}y\left(  d+\right)  , \label{i2}%
\end{align}
where $h_{1},h_{2},\eta>0,\  \lambda$ is a complex eigenvalue parameter, $v$ is
a real-valued continuous functions on $[0,d)\cup(d,q^{-1}a]$ and has finite
limits $v(d\pm),$ $v\left(  \zeta \right)  \geq0,\  \zeta \in \lbrack
0,d)\cup(d,q^{-1}a],$ $a>0.$

$H=L_{q}^{2}(0,d)\overset{\cdot}{+}L_{q}^{2}\left(  d,a\right)  $ is a Hilbert
space endowed with the following inner product
\[
\mathcal{\langle}y,z\rangle:=\int_{0}^{d}y^{\left(  1\right)  }\overline
{z^{\left(  1\right)  }}d_{q}\zeta+\int_{d}^{a}y^{\left(  2\right)  }%
\overline{z^{\left(  2\right)  }}d_{q}\zeta,
\]
where
\[
y(\zeta)=\left \{
\begin{array}
	[c]{cc}%
	y^{\left(  1\right)  }(\zeta), & \zeta \in(0,d)\\
	y^{\left(  2\right)  }(\zeta), & \zeta \in(d,a)
\end{array}
\right.
\]
and%
\[
z(\zeta)=\left \{
\begin{array}
	[c]{cc}%
	z^{\left(  1\right)  }(\zeta), & \zeta \in(0,d)\\
	z^{\left(  2\right)  }(\zeta), & \zeta \in(d,a).
\end{array}
\right.
\]


Let%
\[
\mathcal{T}:\mathcal{D}\subset H\rightarrow H,\text{ }\mathcal{T}y=\tau
y,\ y\in \mathcal{D},
\]
where
\[
\mathcal{D}=\left \{  y\in H:%
\begin{array}
	[c]{c}%
	y\  \text{and\ }D_{q}y\  \text{are }q\text{-regular at zero,}\\
	y\left(  d\pm \right)  \text{\ and\ }\mathcal{D}_{q^{-1}}y\left(  d\pm \right)
	\  \text{exist, }\\
	y\left(  0\right)  -h_{1}D_{q^{-1}}y\left(  0\right)  =0,\\
	y\left(  a\right)  +h_{2}D_{q^{-1}}y\left(  a\right)  =0,\\
	y\left(  d-\right)  =\eta y\left(  d+\right)  ,\\
	\mathcal{D}_{q^{-1}}y\left(  d-\right)  =\frac{1}{\eta}\mathcal{D}_{q^{-1}%
	}y\left(  d+\right)  ,\text{ and}\\
	\tau y\in H
\end{array}
\right \}  .
\]


Let $y,z\in \mathcal{D}.\ $Then we have%
\[
\int_{0}^{a}\left[  (\tau y)(x)\overline{z(x)}-y(x)\overline{(\tau
	z)(x)}\right]  d_{q}x
\]%
\begin{equation}
	=[y,z]\left(  a\right)  -[y,z]\left(  d+\right)  +[y,z]\left(  d-\right)
	-[y,z]\left(  0\right)  , \label{lag}%
\end{equation}
where
\[
\lbrack y,z]:=y(\overline{D_{q^{-1}}z})-(D_{q^{-1}}y)\overline{z}.\
\]


\begin{theorem}
	$\mathcal{T}$\textit{\ is a\ positive self-adjoint operator in }$H$\textit{.}
\end{theorem}

\begin{proof}
	Let $y,z\in \mathcal{D}.\ $It follows from conditions (\ref{1.3})-(\ref{i2})
	and (\ref{lag}) that%
	\begin{equation}
		\langle \mathcal{T}y,z\rangle=\langle y,\mathcal{T}z\rangle. \label{s}%
	\end{equation}
	Since $\mathcal{D}$ is a dense subset in $H,$ we see that $\mathcal{T}%
	$\textit{\ is a\ self-adjoint operator.}
	
	Let
	\[
	y(\zeta)=\left \{
	\begin{array}
		[c]{cc}%
		y^{\left(  1\right)  }(\zeta), & \zeta \in(0,d)\\
		y^{\left(  2\right)  }(\zeta), & \zeta \in(d,a),
	\end{array}
	\right.  y\in \mathcal{D}.
	\]
	Then we get%
	\[
	\langle \mathcal{T}y,y\rangle=\int_{0}^{d}\left[  -\frac{1}{q}D_{q^{-1}}%
	D_{q}y^{\left(  1\right)  }\left(  \zeta \right)  +v\left(  \zeta \right)
	y^{\left(  1\right)  }\left(  \zeta \right)  \right]  \overline{y^{\left(
			1\right)  }\left(  \zeta \right)  }d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}\left[  -\frac{1}{q}D_{q^{-1}}D_{q}y^{\left(  2\right)  }\left(
	\zeta \right)  +v\left(  \zeta \right)  y^{\left(  2\right)  }\left(
	\zeta \right)  \right]  \overline{y^{\left(  2\right)  }\left(  \zeta \right)
	}d_{q}\zeta
	\]%
	\[
	=\int_{0}^{d}-\frac{1}{q}\left(  D_{q^{-1}}D_{q}y^{\left(  1\right)  }\left(
	\zeta \right)  \right)  \overline{y^{\left(  1\right)  }\left(  \zeta \right)
	}d_{q}\zeta+\int_{0}^{d}v\left(  \zeta \right)  \left \vert y^{\left(  1\right)
	}\left(  \zeta \right)  \right \vert ^{2}d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}-\frac{1}{q}\left(  D_{q^{-1}}D_{q}y^{\left(  2\right)  }\left(
	\zeta \right)  \right)  \overline{y^{\left(  2\right)  }\left(  \zeta \right)
	}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)  \left \vert y^{\left(  2\right)
	}\left(  \zeta \right)  \right \vert ^{2}d_{q}\zeta
	\]%
	\[
	=-\frac{1}{q}\int_{0}^{d}D_{q}\left[  D_{q}y^{\left(  1\right)  }\left(
	q^{-1}\zeta \right)  \right]  \overline{y\left(  \zeta \right)  }d_{q}\zeta
	+\int_{0}^{d}v\left(  \zeta \right)  \left \vert y^{\left(  1\right)  }\left(
	\zeta \right)  \right \vert ^{2}d_{q}\zeta
	\]%
	\[
	-\int_{d}^{a}D_{q}\left[  D_{q}y^{\left(  2\right)  }\left(  q^{-1}%
	\zeta \right)  \right]  \overline{y^{\left(  2\right)  }\left(  \zeta \right)
	}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)  \left \vert y^{\left(  2\right)
	}\left(  \zeta \right)  \right \vert ^{2}d_{q}\zeta
	\]%
	\begin{align*}
		&  =-\frac{1}{q}\left[  D_{q}y^{\left(  1\right)  }\left(  q^{-1}d-\right)
		\overline{y^{\left(  1\right)  }\left(  d-\right)  }\right]  +\frac{1}%
		{q}\left[  D_{q}y^{\left(  1\right)  }\left(  0\right)  \overline{y^{\left(
				1\right)  }\left(  0\right)  }\right] \\
		& \\
		&  -\frac{1}{q}\left[  D_{q}y^{\left(  2\right)  }\left(  q^{-1}a\right)
		\overline{y^{\left(  2\right)  }\left(  a\right)  }\right]  +\frac{1}%
		{q}\left[  D_{q}y^{\left(  2\right)  }\left(  d+\right)  \overline{y^{\left(
				2\right)  }\left(  d+\right)  }\right] \\
		& \\
		&  +\frac{1}{q}\int_{0}^{d}\left \vert D_{q}y^{\left(  1\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{0}^{d}v\left(  \zeta \right)
		\left \vert y^{\left(  1\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta \\
		& \\
		&  +\frac{1}{q}\int_{d}^{a}\left \vert D_{q}y^{\left(  2\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)
		\left \vert y^{\left(  2\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta
	\end{align*}%
	\begin{align*}
		&  =\frac{1}{q}h_{1}\left \vert D_{q}y^{\left(  1\right)  }\left(  0\right)
		\right \vert ^{2}+\frac{1}{q}h_{2}\left \vert D_{q}y^{\left(  2\right)  }\left(
		q^{-1}a\right)  \right \vert ^{2}\\
		& \\
		&  +\frac{1}{q}\int_{0}^{d}\left \vert D_{q}y^{\left(  1\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{0}^{d}v\left(  \zeta \right)
		\left \vert y^{\left(  1\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta \\
		& \\
		&  +\frac{1}{q}\int_{d}^{a}\left \vert D_{q}y^{\left(  2\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)
		\left \vert y^{\left(  2\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta>0,
	\end{align*}
	due to $h_{1},h_{2}>0$ and $v\left(  \zeta \right)  \geq0$ for $\zeta \in
	\lbrack0,q^{-1}a].$
\end{proof}

Let
\[
u\left(  \zeta \right)  =\left \{
\begin{array}
	[c]{cc}%
	u^{\left(  1\right)  }(\zeta), & \zeta \in(0,d)\\
	u^{\left(  2\right)  }(\zeta), & \zeta \in(d,a)
\end{array}
\right.
\]
$\ $and$\ $%
\[
\chi \left(  \zeta \right)  =\left \{
\begin{array}
	[c]{cc}%
	\chi^{\left(  1\right)  }(\zeta), & \zeta \in(0,d)\\
	\chi^{\left(  2\right)  }(\zeta), & \zeta \in(d,a)
\end{array}
\right.
\]
$\ $be solutions of the problem%
\[
-\frac{1}{q}D_{q^{-1}}D_{q}y\left(  \zeta \right)  +v\left(  \zeta \right)
y\left(  \zeta \right)  =0,
\]
\[
y\left(  d-\right)  =\eta y\left(  d+\right)  ,
\]%
\[
\mathcal{D}_{q^{-1}}y\left(  d-\right)  =\frac{1}{\eta}\mathcal{D}_{q^{-1}%
}y\left(  d+\right)  ,
\]
satisfying
\begin{align*}
	u^{\left(  1\right)  }\left(  0\right)   &  =h_{1},\ D_{q^{-1}}u^{\left(
		1\right)  }\left(  0\right)  =1,\\
	\chi^{\left(  2\right)  }\left(  a\right)   &  =-h_{2},\ D_{q^{-1}}%
	\chi^{\left(  2\right)  }\left(  a\right)  =1.
\end{align*}


\begin{lemma}
	Zero is not an eigenvalue of the operator $\mathcal{T}.$
\end{lemma}

\begin{proof}
	Let $y\in H$ and $\mathcal{T}y=0.$ Then we have%
	\[
	-\frac{1}{q}D_{q^{-1}}D_{q}y\left(  \zeta \right)  +v\left(  \zeta \right)
	y\left(  \zeta \right)  =0,
	\]
	and%
	\[
	y\left(  \zeta \right)  =\left \{
	\begin{array}
		[c]{cc}%
		c_{1}u^{\left(  1\right)  }\left(  \zeta \right)  +c_{2}\chi^{\left(  1\right)
		}\left(  \zeta \right)  , & \zeta \in(0,d)\\
		c_{3}u^{\left(  2\right)  }\left(  \zeta \right)  +c_{4}\chi^{\left(  2\right)
		}\left(  \zeta \right)  , & \zeta \in(d,a),
	\end{array}
	\right.
	\]
	where $c_{1},c_{2},c_{3}$ and $c_{4}$ are constants$.\ $From conditions
	(\ref{1.3})-(\ref{i2}), we infer that $y=0.$
\end{proof}

\begin{definition}
	The $q$-Wronskian of $y~$and$\ z$ is defined as%
	\[
	W_{q}\left(  y,z\right)  :=yD_{q}z-zD_{q}y.
	\]
	
\end{definition}

\begin{theorem}
	Let%
	\begin{equation}
		G\left(  \zeta,t\right)  =-\frac{1}{W_{q}\left(  u,\chi \right)  }\left \{
		\begin{array}
			[c]{cc}%
			u\left(  \zeta \right)  \chi \left(  t\right)  , & 0\leq \zeta \leq t\leq a,\text{
			}\zeta \neq d,\ t\neq d,\\
			u\left(  t\right)  \chi \left(  \zeta \right)  , & 0\leq t\leq \zeta \leq a,\text{
			}\zeta \neq d,\ t\neq d.
		\end{array}
		\right.  \label{29}%
	\end{equation}
	Then $G\left(  \zeta,t\right)  $ is a $q$-Hilbert--Schmidt kernel, i.e.,
	\[
	\int_{0}^{d}\int_{0}^{d}\left \vert G\left(  \zeta,t\right)  \right \vert
	^{2}d_{q}\zeta d_{q}t<\infty,\  \int_{d}^{a}\int_{d}^{a}|G(\zeta,t)|^{2}%
	d_{q}\zeta d_{q}t<\infty.
	\]
	
\end{theorem}

\begin{proof}
	By (\ref{29}), we see that
	\[
	\int_{0}^{d}d_{q}\zeta \int_{0}^{d}\left \vert G\left(  \zeta,t\right)
	\right \vert ^{2}d_{q}t<\infty,
	\]%
	\[
	\int_{d}^{a}d_{q}\zeta \int_{d}^{a}\left \vert G\left(  \zeta,t\right)
	\right \vert ^{2}d_{q}t<\infty
	\]
	due to $u\left(  .\right)  \chi \left(  .\right)  \in H\times H.$ Then, we
	obtain%
	\begin{equation}
		\int_{0}^{d}\int_{0}^{d}\left \vert G\left(  \zeta,t\right)  \right \vert
		^{2}d_{q}\zeta d_{q}t<\infty,\  \int_{d}^{a}\int_{d}^{a}|G(\zeta,t)|^{2}%
		d_{q}\zeta d_{q}t<\infty. \label{31}%
	\end{equation}
	
\end{proof}

\begin{theorem}
	[\cite{nai}]Let $A$ be an operator defined as
	\[
	A\{ \zeta_{i}\}=\{y_{i}\},
	\]
	where $i\in \mathbb{N}:=\{1,2,3,...\}$ and%
	\begin{equation}
		y_{i}=\sum_{k=1}^{\infty}a_{ik}\zeta_{k}. \label{30}%
	\end{equation}
	If
	\begin{equation}
		\sum_{i,k=1}^{\infty}\left \vert a_{ik}\right \vert ^{2}<\infty, \label{28}%
	\end{equation}
	then $A$ is compact in $l^{2}.$
\end{theorem}

\begin{theorem}
	Let $K:H\rightarrow H$ be an operator defined as
	\begin{equation}
		(Kf)\left(  \zeta \right)  =\left \{
		\begin{array}
			[c]{cc}%
			\int_{0}^{d}G\left(  \zeta,\gamma \right)  f^{\left(  1\right)  }(\gamma
			)d_{q}\gamma, & \zeta \in \lbrack0,d)\\
			\int_{d}^{a}G\left(  \zeta,\gamma \right)  f^{\left(  2\right)  }(\gamma
			)d_{q}\gamma, & \zeta \in(d,a],
		\end{array}
		\right.  \label{300}%
	\end{equation}
	where%
	\[
	f(\zeta)=\left \{
	\begin{array}
		[c]{cc}%
		f^{\left(  1\right)  }(\zeta), & \zeta \in \lbrack0,d)\\
		f^{\left(  2\right)  }(\zeta), & \zeta \in(d,a],
	\end{array}
	\right.  f\in H.
	\]
	Then $K$ is a compact operator.
\end{theorem}

\begin{proof}
	Let
	\[
	\phi_{i}=\phi_{i}\left(  \zeta \right)  =\left \{
	\begin{array}
		[c]{cc}%
		\phi_{i}^{\left(  1\right)  }(\zeta), & \zeta \in \lbrack0,d)\\
		\phi_{i}^{\left(  2\right)  }(\zeta), & \zeta \in(d,a]
	\end{array}
	\right.  \ (i\in \mathbb{N)}%
	\]
	be a complete, orthonormal basis of $H.\ $Let$\ i,k\in \mathbb{N}.$ If we set%
	\[
	\zeta_{i}=\langle f,\phi_{i}\rangle=\int_{0}^{d}f^{\left(  1\right)  }\left(
	\zeta \right)  \overline{\phi_{i}^{\left(  1\right)  }\left(  \zeta \right)
	}d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}f^{\left(  2\right)  }\left(  \zeta \right)  \overline{\phi
		_{i}^{\left(  2\right)  }\left(  \zeta \right)  }d_{q}\zeta,
	\]%
	\[
	y_{i}=\langle g,\phi_{i}\rangle=\int_{0}^{d}g^{\left(  1\right)  }\left(
	\zeta \right)  \overline{\phi_{i}^{\left(  1\right)  }\left(  \zeta \right)
	}d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}g^{\left(  2\right)  }\left(  \zeta \right)  \overline{\phi
		_{i}^{\left(  2\right)  }\left(  \zeta \right)  }d_{q}\zeta,
	\]%
	\[
	a_{ik}=\int_{0}^{d}\int_{0}^{d}G\left(  \zeta,t\right)  \overline{\phi
		_{i}^{\left(  1\right)  }\left(  \zeta \right)  }\overline{\phi_{k}^{\left(
			1\right)  }\left(  t\right)  }d_{q}\zeta d_{q}t\text{ }%
	\]%
	\[
	+\int_{d}^{a}\int_{d}^{a}G\left(  \zeta,t\right)  \overline{\phi_{i}^{\left(
			2\right)  }\left(  \zeta \right)  }\overline{\phi_{k}^{\left(  2\right)
		}\left(  t\right)  }d_{q}\zeta d_{q}t.
	\]
	Then, $H\ $ is mapped isometrically $l^{2}.$ Then, $K$ transforms into $A$
	defined as (\ref{30}) in $l^{2}\ $and (\ref{31}) is translated into
	(\ref{28}). From Theorem 3, $A$ is compact. Thus, $K$ is compact.
\end{proof}

Since$\ K=\mathcal{T}^{-1},\ $the completeness of the system of all
eigenvectors of $\mathcal{T}$ is equivalent to the completeness of the system
of all eigenvectors of $K$. By the Hilbert--Schmidt theorem, we obtain the
following theorem.

\begin{theorem}
	For the boundary-value problem (\ref{1.1})-(\ref{i2}), there exists an
	orthonormal basis $\left \{  \psi_{k}\right \}  _{k\in \mathbb{N}}\ $in $H.$ For
	$f\in H,$ we have%
	\begin{equation}
		f\left(  \zeta \right)  =\sum_{k=1}^{\infty}c_{k}\psi_{k}\left(  \zeta \right)
		, \label{ser}%
	\end{equation}
	where
	\[
	c_{k}=\langle f,\psi_{k}\rangle,\text{ }k\in \mathbb{N}.
	\]
	Thus, we get
	\begin{equation}
		\lim_{N\rightarrow \infty}\left \{
		\begin{array}
			[c]{c}%
			\int_{0}^{d}\left \vert f^{\left(  1\right)  }\left(  \zeta \right)  -\sum
			_{k=1}^{N}c_{k}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  \right \vert
			^{2}d_{q}\zeta \\
			+\int_{d}^{a}\left \vert f^{\left(  2\right)  }\left(  \zeta \right)
			-\sum_{k=1}^{N}c_{k}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)
			\right \vert ^{2}d_{q}\zeta
		\end{array}
		\right \}  =0, \label{38}%
	\end{equation}
	Moreover, it follows from (\ref{38}) that
	\begin{equation}
		\int_{0}^{d}\left \vert f^{\left(  1\right)  }\left(  \zeta \right)  \right \vert
		^{2}d_{q}\zeta+\int_{d}^{a}\left \vert f^{\left(  2\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta=\sum_{k=1}^{\infty}\left \vert
		c_{k}\right \vert ^{2}. \label{39}%
	\end{equation}
	
\end{theorem}

Now let's prove the main result of the article

\begin{theorem}
	Let $f,D_{q}f:[0,q^{-1}a]\rightarrow \mathbb{R}$ be continuous functions on
	$[0,d)\cup(d,q^{-1}a],$ has finite limits $f(d\pm),$ $D_{q^{-1}}f(d\pm)$ and
	satisfying conditions (\ref{1.3})-(\ref{i2}). Then the series
	\begin{equation}
		f\left(  \zeta \right)  =\sum_{k=1}^{\infty}c_{k}\psi_{k}\left(  \zeta \right)
		, \label{40}%
	\end{equation}
	where
	\[
	c_{k}=\langle f,\psi_{k}\rangle,\text{ }k\in \mathbb{N},
	\]
	converges uniformly to $f\ $on the set $[0,d)\cup(d,a].$
\end{theorem}

\begin{proof}
	Let
	\begin{align}
		S\left(  y\right)   &  :=\frac{1}{q}h_{1}\left \vert D_{q}y^{\left(  1\right)
		}\left(  0\right)  \right \vert ^{2}+\frac{1}{q}h_{2}\left \vert D_{q}y^{\left(
			2\right)  }\left(  q^{-1}a\right)  \right \vert ^{2}\nonumber \\
		& \nonumber \\
		&  +\frac{1}{q}\int_{0}^{d}\left \vert D_{q}y^{\left(  1\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{0}^{d}v\left(  \zeta \right)
		\left \vert y^{\left(  1\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta \nonumber \\
		& \nonumber \\
		&  +\frac{1}{q}\int_{d}^{a}\left \vert D_{q}y^{\left(  2\right)  }\left(
		\zeta \right)  \right \vert ^{2}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)
		\left \vert y^{\left(  2\right)  }\left(  \zeta \right)  \right \vert ^{2}%
		d_{q}\zeta, \label{func}%
	\end{align}
	and $S(y)\geq0$. If we take%
	\[
	y=f\left(  \zeta \right)  -\sum_{k=1}^{N}c_{k}\psi_{k}\left(  \zeta \right)
	\]
	in (\ref{func}), we conclude that%
	\[
	S\left(  f\left(  \zeta \right)  -\sum_{k=1}^{N}c_{k}\psi_{k}\left(
	\zeta \right)  \right)
	\]%
	\[
	=\frac{1}{q}h_{1}\left[  D_{q}f^{\left(  1\right)  }\left(  0\right)
	-\sum_{k=1}^{N}c_{k}D_{q}\psi_{k}^{\left(  1\right)  }\left(  0\right)
	\right]  ^{2}%
	\]%
	\[
	+\frac{1}{q}h_{2}\left[  D_{q}f^{\left(  2\right)  }\left(  q^{-1}a\right)
	-\sum_{k=1}^{N}c_{k}\left(  D_{q}\psi_{k}^{\left(  2\right)  }\left(
	a\right)  \right)  \right]  ^{2}%
	\]%
	\[
	+\frac{1}{q}\int_{0}^{d}\left(  D_{q}f^{\left(  1\right)  }\left(
	\zeta \right)  -\sum_{k=1}^{N}c_{k}D_{q}\psi_{k}^{\left(  1\right)  }\left(
	\zeta \right)  \right)  ^{2}d_{q}\zeta
	\]%
	\[
	+\frac{1}{q}\int_{d}^{a}\left(  D_{q}f^{\left(  2\right)  }\left(
	\zeta \right)  -\sum_{k=1}^{N}c_{k}D_{q}\psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  \right)  ^{2}d_{q}\zeta
	\]%
	\begin{align*}
		&  +\int_{0}^{a}v\left(  \zeta \right)  \left(  f^{\left(  1\right)  }\left(
		\zeta \right)  -\sum_{k=1}^{N}c_{k}\psi_{k}^{\left(  1\right)  }\left(
		\zeta \right)  \right)  ^{2}d_{q}\zeta \\
		& \\
		&  +\int_{d}^{a}v\left(  \zeta \right)  \left(  f^{\left(  2\right)  }\left(
		\zeta \right)  -\sum_{k=1}^{N}c_{k}\psi_{k}^{\left(  2\right)  }\left(
		\zeta \right)  \right)  ^{2}d_{q}\zeta
	\end{align*}%
	\[
	=\frac{1}{q}h_{1}\left[  D_{q}f^{\left(  1\right)  }\left(  0\right)  \right]
	^{2}+\frac{1}{q}h_{2}\left[  D_{q}f^{\left(  2\right)  }\left(  q^{-1}%
	a\right)  \right]  ^{2}%
	\]%
	\[
	-2\frac{1}{q}\sum_{k=1}^{N}c_{k}\left[
	\begin{array}
		[c]{c}%
		-h_{1}D_{q}f^{\left(  1\right)  }\left(  0\right)  D_{q}\psi_{k}^{\left(
			1\right)  }\left(  0\right) \\
		\\
		-h_{2}D_{q}f^{\left(  2\right)  }\left(  q^{-1}a\right)  D_{q}\psi
		_{k}^{\left(  2\right)  }\left(  a\right)
	\end{array}
	\right]
	\]%
	\[
	-\frac{1}{q}\sum_{k,m=1}^{N}c_{k}c_{m}\left[
	\begin{array}
		[c]{c}%
		-h_{1}D_{q}\psi_{k}^{\left(  1\right)  }\left(  0\right)  D_{q}\psi
		_{m}^{\left(  1\right)  }\left(  0\right) \\
		\\
		-h_{2}D_{q}\psi_{k}^{\left(  2\right)  }\left(  a\right)  D_{q}\psi
		_{m}^{\left(  2\right)  }\left(  a\right)
	\end{array}
	\right]
	\]%
	\begin{align*}
		&  +\frac{1}{q}\int_{0}^{d}(D_{q}f^{\left(  1\right)  }\left(  \zeta \right)
		)^{2}d_{q}\zeta+\int_{0}^{d}v\left(  \zeta \right)  f^{\left(  1\right)
			2}\left(  \zeta \right)  d_{q}\zeta \\
		& \\
		&  +\frac{1}{q}\int_{d}^{a}(D_{q}f^{\left(  2\right)  }\left(  \zeta \right)
		)^{2}d_{q}\zeta+\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)
			2}\left(  \zeta \right)  d_{q}\zeta
	\end{align*}%
	\[
	-2\frac{1}{q}\sum_{k=1}^{N}c_{k}\left[
	\begin{array}
		[c]{c}%
		\int_{0}^{d}D_{q}f^{\left(  1\right)  }\left(  \zeta \right)  D_{q}\psi
		_{k}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta \\
		+\int_{d}^{a}D_{q}f^{\left(  2\right)  }\left(  \zeta \right)  D_{q}\psi
		_{k}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\end{array}
	\right]
	\]%
	\[
	-2\sum_{k=1}^{N}c_{k}\left[
	\begin{array}
		[c]{c}%
		\int_{0}^{d}v\left(  \zeta \right)  f^{\left(  1\right)  }\left(  \zeta \right)
		\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta \\
		+\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)  }\left(
		\zeta \right)  \psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\end{array}
	\right]
	\]%
	\[
	+\frac{1}{q}\sum_{k,m=1}^{N}c_{k}c_{m}\left[
	\begin{array}
		[c]{c}%
		\int_{0}^{d}D_{q}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  D_{q}%
		\psi_{m}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta \\
		+\int_{d}^{a}D_{q}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  D_{q}%
		\psi_{m}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\end{array}
	\right]
	\]%
	\[
	+\sum_{k,m=1}^{N}c_{k}c_{m}\left[
	\begin{array}
		[c]{c}%
		\int_{0}^{d}v\left(  \zeta \right)  \psi_{k}^{\left(  1\right)  }\left(
		\zeta \right)  \psi_{m}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta \\
		+\int_{d}^{a}v\left(  \zeta \right)  \psi_{k}^{\left(  2\right)  }\left(
		\zeta \right)  \psi_{m}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\end{array}
	\right]  .
	\]
	Applications of (\ref{1.3})-(\ref{i2}) and $q$-integration by parts give%
	\[
	\frac{1}{q}\int_{0}^{d}D_{q}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)
	D_{q}f^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta+\int_{0}%
	^{d}v\left(  \zeta \right)  f^{\left(  1\right)  }\left(  \zeta \right)
	\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta
	\]%
	\[
	+\frac{1}{q}\int_{d}^{a}D_{q}\psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  D_{q}f^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	+\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)  }\left(
	\zeta \right)  \psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\]%
	\[
	=\frac{1}{q}D_{q}\psi_{k}^{\left(  1\right)  }\left(  q^{-1}d-\right)
	f^{\left(  1\right)  }\left(  d-\right)  -\frac{1}{q}D_{q}\psi_{k}^{\left(
		1\right)  }\left(  0\right)  f^{\left(  1\right)  }\left(  0\right)
	\]%
	\[
	+\frac{1}{q}D_{q}\psi_{k}^{\left(  2\right)  }\left(  q^{-1}a\right)
	f^{\left(  2\right)  }\left(  a\right)  -\frac{1}{q}D_{q}\psi_{k}^{\left(
		2\right)  }\left(  d+\right)  f^{\left(  2\right)  }\left(  d+\right)
	\]%
	\[
	-\frac{1}{q}\int_{0}^{d}f^{\left(  1\right)  }\left(  \zeta \right)  D_{q^{-1}%
	}\left(  D_{q}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  \right)
	d_{q}\zeta
	\]%
	\[
	-\frac{1}{q}\int_{d}^{a}f^{\left(  2\right)  }\left(  \zeta \right)  D_{q^{-1}%
	}\left(  D_{q}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  \right)
	d_{q}\zeta
	\]%
	\[
	+\int_{0}^{d}v\left(  \zeta \right)  f^{\left(  1\right)  }\left(
	\zeta \right)  \psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}%
	\zeta+\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)  }\left(
	\zeta \right)  \psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\]%
	\[
	=-\frac{1}{q}h_{2}D_{q^{-1}}f^{\left(  2\right)  }\left(  a\right)  D_{q^{-1}%
	}\psi_{k}^{\left(  2\right)  }\left(  a\right)
	\]%
	\[
	-\frac{1}{q}h_{1}D_{q^{-1}}f^{\left(  1\right)  }\left(  0\right)  D_{q}%
	\psi_{k}^{\left(  1\right)  }\left(  0\right)
	\]%
	\[
	+\int_{0}^{d}f^{\left(  1\right)  }\left(  \zeta \right)  \left[  -\frac{1}%
	{q}D_{q^{-1}}\left(  D_{q}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)
	\right)  +v\left(  \zeta \right)  \psi_{k}^{\left(  1\right)  }\left(
	\zeta \right)  \right]  d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}f^{\left(  2\right)  }\left(  \zeta \right)  \left[  -\frac{1}%
	{q}D_{q^{-1}}\left(  D_{q}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)
	\right)  +v\left(  \zeta \right)  \psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  \right]  d_{q}\zeta
	\]%
	\[
	=-h_{2}\frac{1}{q}D_{q^{-1}}f^{\left(  2\right)  }\left(  a\right)  D_{q^{-1}%
	}\psi_{k}^{\left(  2\right)  }\left(  a\right)
	\]%
	\[
	-h_{1}\frac{1}{q}D_{q^{-1}}f^{\left(  1\right)  }\left(  0\right)  D_{q}%
	\psi_{k}^{\left(  1\right)  }\left(  0\right)  +\lambda_{k}c_{k},
	\]
	and%
	\[
	\frac{1}{q}\int_{0}^{d}D_{q}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)
	D_{q}\psi_{m}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}\zeta
	\]%
	\[
	+\frac{1}{q}\int_{d}^{a}D_{q}\psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  D_{q}\psi_{m}^{\left(  2\right)  }\left(  \zeta \right)
	d_{q}\zeta
	\]%
	\[
	+\int_{0}^{d}v\left(  \zeta \right)  \psi_{k}^{\left(  1\right)  }\left(
	\zeta \right)  \psi_{m}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}%
	\zeta+\int_{d}^{a}v\left(  \zeta \right)  \psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  \psi_{m}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta
	\]%
	\[
	=\frac{1}{q}D_{q}\psi_{m}^{\left(  1\right)  }\left(  q^{-1}d-\right)
	\psi_{k}^{\left(  1\right)  }\left(  d-\right)  +\frac{1}{q}D_{q}\psi
	_{m}^{\left(  2\right)  }\left(  q^{-1}a\right)  \psi_{k}^{\left(  2\right)
	}\left(  a\right)
	\]%
	\[
	-\frac{1}{q}D_{q}\psi_{m}^{\left(  1\right)  }\left(  0\right)  \psi
	_{k}^{\left(  1\right)  }\left(  0\right)  -\frac{1}{q}D_{q}\psi_{m}^{\left(
		2\right)  }\left(  d+\right)  \psi_{k}^{\left(  2\right)  }\left(  d+\right)
	\]%
	\[
	+\int_{0}^{d}\psi_{k}^{\left(  1\right)  }\left(  \zeta \right)  \left[
	-\frac{1}{q}D_{q^{-1}}\left(  D_{q}\psi_{m}^{\left(  1\right)  }\left(
	\zeta \right)  \right)  +v\left(  \zeta \right)  \psi_{m}^{\left(  1\right)
	}\left(  \zeta \right)  \right]  d_{q}\zeta
	\]%
	\[
	+\int_{d}^{a}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  \left[
	-\frac{1}{q}D_{q^{-1}}\left(  D_{q}\psi_{m}^{\left(  2\right)  }\left(
	\zeta \right)  \right)  +v\left(  \zeta \right)  \psi_{m}^{\left(  2\right)
	}\left(  \zeta \right)  \right]  d_{q}\zeta
	\]%
	\[
	=\frac{1}{q}\psi_{k}^{\left(  2\right)  }\left(  a\right)  D_{q^{-1}}\psi
	_{m}^{\left(  2\right)  }\left(  a\right)  -\frac{1}{q}\psi_{k}^{\left(
		1\right)  }\left(  0\right)  D_{q^{-1}}\psi_{k}^{\left(  1\right)  }\left(
	0\right)
	\]%
	\[
	+\lambda_{k}\left[  \int_{0}^{d}\psi_{k}^{\left(  1\right)  }\left(
	\zeta \right)  \psi_{m}^{\left(  1\right)  }\left(  \zeta \right)  d_{q}%
	\zeta+\int_{d}^{a}\psi_{k}^{\left(  2\right)  }\left(  \zeta \right)  \psi
	_{m}^{\left(  2\right)  }\left(  \zeta \right)  d_{q}\zeta \right]
	\]%
	\[
	=-\frac{1}{q}h_{1}D_{q^{-1}}\psi_{k}^{\left(  2\right)  }\left(  a\right)
	D_{q^{-1}}\psi_{m}^{\left(  2\right)  }\left(  a\right)
	\]%
	\[
	-\frac{1}{q}h_{2}D_{q^{-1}}\psi_{k}^{\left(  1\right)  }\left(  0\right)
	D_{q^{-1}}\psi_{m}^{\left(  1\right)  }\left(  0\right)  +\lambda_{k}%
	\delta_{km},
	\]
	where
	\[
	\delta_{km}:=\left \{
	\begin{array}
		[c]{cc}%
		1, & \text{if\ }k=m\\
		0, & \text{if }k\neq m.
	\end{array}
	\right.
	\]
	Hence%
	\begin{align*}
		&  S\left(  f\left(  \zeta \right)  -\sum_{k=1}^{N}c_{k}\psi_{k}\left(
		\zeta \right)  \right)  =\frac{1}{q}h_{1}\left[  D_{q}f^{\left(  1\right)
		}\left(  0\right)  \right]  ^{2}\\
		& \\
		&  +\frac{1}{q}h_{2}\left[  D_{q}f^{\left(  2\right)  }\left(  q^{-1}a\right)
		\right]  ^{2}+\frac{1}{q}\int_{0}^{d}(D_{q}f^{\left(  1\right)  }\left(
		\zeta \right)  )^{2}d_{q}\zeta \\
		& \\
		&  +\int_{0}^{d}v\left(  \zeta \right)  f^{\left(  1\right)  2}\left(
		\zeta \right)  d_{q}\zeta+\frac{1}{q}\int_{d}^{a}(D_{q}f^{\left(  2\right)
		}\left(  \zeta \right)  )^{2}d_{q}\zeta \\
		& \\
		&  +\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)  2}\left(
		\zeta \right)  d_{q}\zeta-\frac{1}{q}\sum_{k=1}^{N}\lambda_{k}c_{k}^{2}.
	\end{align*}
	Then we get%
	\[
	\sum_{k=1}^{\infty}\lambda_{k}c_{k}^{2}\leq h_{1}\left[  D_{q}f^{\left(
		1\right)  }\left(  0\right)  \right]  ^{2}+h_{2}\left[  D_{q}f^{\left(
		2\right)  }\left(  q^{-1}a\right)  \right]  ^{2}%
	\]%
	\begin{align}
		&  +\int_{0}^{d}(D_{q}f^{\left(  1\right)  }\left(  \zeta \right)  )^{2}%
		d_{q}\zeta+q\int_{0}^{d}v\left(  \zeta \right)  f^{\left(  1\right)  2}\left(
		\zeta \right)  d_{q}\zeta \nonumber \\
		& \nonumber \\
		&  +\int_{d}^{a}(D_{q}f^{\left(  2\right)  }\left(  \zeta \right)  )^{2}%
		d_{q}\zeta+q\int_{d}^{a}v\left(  \zeta \right)  f^{\left(  2\right)  2}\left(
		\zeta \right)  d_{q}\zeta. \label{bes}%
	\end{align}
	due to $S$ is nonnegative for all $N$.\ Thus, the convergence of the series
	\[
	\sum_{k=1}^{\infty}\lambda_{k}c_{k}^{2}%
	\]
	follows.
	
	Now,\ we shall prove that the series
	\begin{equation}
		\sum_{k=1}^{\infty}\left \vert c_{k}\psi_{k}\left(  \zeta \right)  \right \vert
		\label{63}%
	\end{equation}
	is uniformly convergent on $[0,d)\cup(d,a].$ Since $\mathcal{T}\psi
	_{k}=\lambda_{k}\psi_{k},$ $k\in \mathbb{N},$ we have
	\[
	\psi_{k}\left(  \zeta \right)  =\lambda_{k}(\mathcal{T}^{-1}\psi_{k})\left(
	\zeta \right)  =\lambda_{k}\langle G\left(  \zeta,t\right)  ,\psi_{k}%
	\rangle,\text{ }k\in \mathbb{N}.
	\]
	If we rewrite the series (\ref{63}), we conclude that%
	\begin{equation}
		\sum_{k=1}^{\infty}\left \vert c_{k}\psi_{k}\left(  \zeta \right)  \right \vert
		=\sum_{k=1}^{\infty}\lambda_{k}\left \vert c_{k}\Upsilon_{k}\left(
		\zeta \right)  \right \vert , \label{65}%
	\end{equation}
	where%
	\[
	\Upsilon_{k}\left(  \zeta \right)  =\langle G\left(  \zeta,t\right)  ,\psi
	_{k}\rangle,\text{ }k\in \mathbb{N}.
	\]
	This can be regarded as the Fourier coefficients of $G\left(  \zeta,t\right)
	$ as a function of $t.$ From (\ref{bes}), we find
	\[
	\sum_{k=1}^{\infty}\lambda_{k}\Upsilon_{k}^{2}\left(  \zeta \right)  \leq
	h_{1}\left[  D_{q}G^{\left(  1\right)  }\left(  \zeta,0\right)  \right]
	^{2}+h_{2}\left[  D_{q}G^{\left(  2\right)  }\left(  \zeta,q^{-1}a\right)
	\right]  ^{2}%
	\]%
	\[
	+\int_{0}^{d}(D_{q}G^{\left(  1\right)  }\left(  \zeta,t\right)  )^{2}%
	d_{q}t+q\int_{0}^{d}v\left(  t\right)  G^{\left(  1\right)  2}\left(
	\zeta,t\right)  d_{q}t
	\]%
	\[
	+\int_{d}^{a}(D_{q}G^{\left(  2\right)  }\left(  \zeta,t\right)  )^{2}%
	d_{q}t+q\int_{0}^{a}v\left(  t\right)  G^{\left(  2\right)  2}\left(
	\zeta,t\right)  d_{q}t.
	\]
	Since all the functions appearing under the integral sign are bounded, we
	deduce that%
	\[
	\sum_{k=1}^{\infty}\lambda_{k}\Upsilon_{k}^{2}\left(  \zeta \right)  \leq C,
	\]
	where $\ C$ is a constant. Applying the Cauchy--Schwartz inequality to the
	series (\ref{65}), we see that%
	\begin{align}
		\sum_{k=n}^{n+m}\lambda_{k}\left \vert c_{k}\Upsilon_{k}\left(  \zeta \right)
		\right \vert  &  \leq \sqrt{\sum_{k=n}^{n+m}\lambda_{k}c_{k}^{2}}\sqrt
		{\sum_{k=n}^{n+m}\lambda_{k}\Upsilon_{k}^{2}\left(  \zeta \right)  }\nonumber \\
		& \nonumber \\
		&  \leq \sqrt{C}\sqrt{\sum_{k=n}^{n+m}\lambda_{k}c_{k}^{2}}. \label{67}%
	\end{align}
	From (\ref{bes}) and (\ref{67}), the series (\ref{63})\ is uniformly
	convergent on $[0,d)\cup(d,a].$ Since
	\[
	\left \vert \sum_{k=1}^{\infty}c_{k}\psi_{k}\left(  \zeta \right)  \right \vert
	\leq \sum_{k=1}^{\infty}\left \vert c_{k}\psi_{k}\left(  \zeta \right)
	\right \vert ,
	\]
	the series (\ref{40}) is also uniformly convergent on$\ [0,d)\cup(d,a].$
	
	Let
	\begin{equation}
		f_{1}\left(  \zeta \right)  =\sum_{k=1}^{\infty}c_{k}\psi_{k}\left(
		\zeta \right)  . \label{68}%
	\end{equation}
	Since the series (\ref{68}) is uniformly convergent on $[0,d)\cup(d,a],$ we
	get%
	\[
	\int_{0}^{d}f_{1}^{\left(  1\right)  }\left(  \zeta \right)  \psi_{k}^{\left(
		1\right)  }\left(  \zeta \right)  d_{q}\zeta+\int_{d}^{a}f_{1}^{\left(
		2\right)  }\left(  \zeta \right)  \psi_{k}^{\left(  2\right)  }\left(
	\zeta \right)  d_{q}\zeta=c_{k}\text{ }(k\in \mathbb{N}).
	\]
	Consequently, the Fourier coefficients of $f$ and $f_{1}$ are the same.
	Applying the Parseval equality (\ref{39}) to the function $f-f_{1},$\ we find
	$f-f_{1}=0,$ due to the Fourier coefficients of the function $f-f_{1}$ are
	zero. This finishes the proof.
\end{proof}


\end{document}
